Author's response to reviews

Title: Impact of early parenteral nutrition completing enteral nutrition in adult critically ill patients (EPaNIC trial): a study protocol and statistical analysis plan for a randomized controlled trial

Authors:

- Michael P Casaer (michael.casaer@uzleuven.be)
- Greet Hermans (greet.hermans@uzleuven.be)
- Alexander Wilmer (alexander.wilmer@uzleuven.be)
- Greet Van den Berghe (greet.vandenberghe@med.kuleuven.be)

Version: 7 Date: 19 January 2011

Author's response to reviews: see over
To the editors, the independent editorial production team and the reviewer,

Dear Sir,

We thank the independent editorial production team for the opportunity to control and correct the final version of our manuscript. We indeed have found some typing errors, they have been corrected and highlighted.

1. Page 10: We replaced the “next morning” by the “second morning in ICU”, the previous wording was ambiguous.
2. Page 11: We added “iMDsoft” (the company) after “Metavision” (the product).
3. Page 15: We replaced “Monitoring will be performed and will be reported following the sponsor’s standing operating procedures” by “Monitoring is performed and reported following the sponsor’s standing operating procedures.”
4. Page 16: We replaced “event” by “events”

We apologize for these errors and thank the editors for their interest in our manuscript and the referees and independent editorial production team for the dedicated review.

Sincerely yours,

Michaël P. Casaer, MD  
Greet Hermans, MD, PhD

Greet Van den Berghe, MD, PhD  
Alexander Wilmer, MD, PhD